

Centre interuniversitaire de recherche sur les réseaux d'entreprise, la logistique et le transport

Interuniversity Research Centre on Enterprise Networks, Logistics and Transportation

# Collaborative Mechanisms for Sustainable Supply Chains: A Systematic Literature Review

Parnian Oktaei Caroline Cloutier Nadia Lehoux

June 2017

**CIRRELT-2017-29** 

Bureaux de Montréal : Université de Montréal Pavillon André-Aisenstadt C.P. 6128, succursale Centre-ville Montréal (Québec) Canada H3C 3J7 Téléphone : 514 343-7575 Télécopie : 514 343-7121 Bureaux de Québec : Université Laval Pavillon Palasis-Prince 2325, de la Terrasse, bureau 2642 Québec (Québec) Canada G1V 0A6 Téléphone: 418 656-2073 Télécopie : 418 656-2624

www.cirrelt.ca





ÉTS UQÀM

HEC MONTRĒAL





# Collaborative Mechanisms for Sustainable Supply Chains: A Systematic Literature Review

# Parnian Oktaei<sup>1</sup>, Caroline Cloutier<sup>1</sup>, Nadia Lehoux<sup>1,2,\*</sup>

- <sup>1</sup> Department of Mechanical Engineering, 1065, avenue de la Médecine, Université Laval, Québec, Canada G1V 0A6
- <sup>2</sup> Interuniversity Research Centre on Enterprise Networks, Logistics and Transportation (CIRRELT)

**Abstract.** Whether from government policies, customer expectations or personal beliefs, there is increasing pressure on firm's and their supply chains to adopt sustainable practices. The concept of sustainability in supply chains has therefore attracted the attention of academia and industry. Most research states the importance of collaboration with upstream and downstream entities as a critical success factor when aiming for a sustainable supply chain and proposes various collaborative mechanisms to enable firms in the implementation of a sustainability initiative. The goal of this paper is to investigate the role of collaboration in these initiatives and explore the proposed collaborative mechanisms via a systematic literature review method. This systematic mapping of the field provides a classification of previous publications, categorizes the collaborative mechanisms, and highlights the role played by these mechanisms when deploying sustainable supply chains.

Keywords. Supply Chains, sustainability, collaboration, systematic literature review.

Results and views expressed in this publication are the sole responsibility of the authors and do not necessarily reflect those of CIRRELT.

Les résultats et opinions contenus dans cette publication ne reflètent pas nécessairement la position du CIRRELT et n'engagent pas sa responsabilité.

- Dépôt légal Bibliothèque et Archives nationales du Québec Bibliothèque et Archives Canada, 2017
- © Oktaei, Cloutier, Lehoux and CIRRELT, 2017

<sup>\*</sup> Corresponding author: Nadia.Lehoux@cirrelt.ca

# 1. Introduction

As the interest for sustainable development gains importance in the public eye, the pressure on businesses to adopt sustainable management practices in their own organisation as well as within their supply chain is increasing. The pursuit for a sustainable supply chain often faces challenges in its attempts to satisfy the triple bottom-line (economic, environmental, and social aspects). Walker *et al.* (2008) identified some of these barriers as cost concerns, lack of training as well as lack of stakeholder or supplier commitment. Businesses which choose to integrate environmental and/or social aspects in their supply chain management will then need to identify enablers to help them overcome these challenges. Walker *et al.* (2008) and Diabat and Govindan (2011) both reviewed the literature in order to identify these enablers. Among all potential enablers of efficiency and sustainability they mention, this research focuses on collaboration as one of the key factors to adopt.

Collaboration has long been recognized as a key factor in traditional supply chain management (Barratt, 2004; Lehoux *et al.*, 2014). Collaboration, as defined by Audy *et al.* (2012), involves two or more parties exchanging or sharing resources in the hopes of creating a synergy that generates benefits greater than what could be achieved individually. In traditional supply chain management (SCM), the benefits considered are usually strictly economic and all partners will readily see the advantages for their business. Green and sustainable SCM entail multi-stakeholder initiatives as well, however, they are slightly different as they include environmental and/or social considerations, sometimes to the detriment of economic benefits. Addressing specific sustainability issues might also require the involvement of new businesses in the supply chain, but all of the entities in the current supply chain as well as the new comers might not necessarily have attained the same level of sustainability. Hence, the importance of aiming for well-defined objectives through collaboration.

Over the years, several researchers have addressed the role of collaborative activities in green or sustainable supply chains such as Vachon and Klassen (2008), Gold *et al.* (2010), Blome *et al.* (2014), Waller *et al.* (2015), and Chin *et al.* (2015). While scholars and practitioners agree on the critical role of collaboration in these supply chains, they do not seem to highlight the collaborative mechanisms that businesses could implement to enable this collaboration such as methods, tools, processes, systems, etc. Furthermore, to the best of our knowledge, no previous research has proposed a comprehensive literature review on the role of collaboration and collaborative mechanisms in sustainable SCM specifically. It should be noted that, in this research, the authors considered all references to collaboration in a green, environmental or sustainable supply chain management context to be relevant to sustainable SCM.

Therefore, this research aims to investigate the usefulness of collaboration in the quest to improve the sustainability of supply chains through a structured literature review (SLR) and to identify the collaborative mechanisms that enable this collaboration. For this SLR, a set of keyword combinations relating to supply chains, collaboration, and sustainability were systemically searched in specific fields of selected search engines. Based on this SLR, more than 190 articles from 37 countries were found and analyzed in order to identify the methodology used as well as the collaborative mechanisms considered in the research. A total of 715 collaborative mechanisms were found in 187 of the articles. A classification exercise led to the identification of seven broad categories of collaborative mechanisms: relationship management, contract and economic practices, joint practices, technology and information sharing practices, governance practices, assessment practices, and supply chain design. Results showed that the top five most frequently mentioned collaborative mechanisms, in decreasing order, are contracts, economic incentives, communication, information sharing, and integration. In almost all of the contributions, authors proposed a combination of collaborative mechanisms from several categories. This article contributes to confirm the usefulness of collaborations in sustainable supply chains while demystifying the role played by specific mechanisms to support the collaboration process.

This paper is structured as follows. Section two presents the methodology used for the structured literature review. Section three addresses how collaboration supports sustainable SCM. Section four describes the collaborative mechanisms found in each category based on the literature review analysis. Section five presents a discussion concerning the mechanisms and their deployment. Section six proposes a conclusion.

# 2. SLR Methodology

The purpose of this research is to investigate the role of collaboration in improving the sustainability of supply chains and to identify the collaborative mechanisms that enable this collaboration. To reach this goal, a systematic literature review analyzing the existing research was conducted. This method allowed us to identify the scope of the review, select a certain number of criteria, and specify a set of questions to answer. Inspired by the approach proposed by Tranfield *et al.* (2003), the following steps were followed. First, a set of research questions were identified. Then, a set of preliminary "Keywords", "Search field", and "Engine" were selected. After performing a trial search and evaluating the preliminary set, the best fit terms and locations for the research were found. Thirdly, a set of papers regarding the predefined criteria and research questions were selected and assessed. Finally, the results were synthesized and descriptive statistics and findings of research proposed.

# 2.1 Research questions

To reach the goal of the research, three key questions had to be addressed:

- Q1- Is collaboration useful to improve the sustainability of supply chains?
- Q2- How did each study come to this conclusion (i.e. methodology followed)?
- Q3- What collaborative mechanisms are proposed to enable this collaboration?

#### 2.2 *Research keywords, search fields, and databases*

In order to define a suitable set of keywords, search fields, and databases, a trial search was done by testing different keywords and search engines. This preliminary step helped us enrich the research by selecting and validating the right criteria that would lead towards a more comprehensive investigation.

Consequently, in this research we investigated a set of electronic databases encompassing: "Google Scholar", "Elsevier Science Direct", "Wiley Online", "Sage Online", "JSTOR", and "Springer link". To narrow the research field and select the most relevant studies, we established 32 triads of keywords that were searched in the following fields of each database: "Title", "Keyword", and "Abstract". The keyword "Supply chain" was in each triad. The second keyword in the triad was either "Collaboration", "Partnership", "Alliance", or "Collaborative". The third keyword in the triad was either "Sustainable", "Sustainability", "Green", "Environmental", "Social", "Economic", "Benefit" or "Financial".

# 2.3 Assessment and selection criteria

In order to make sure that the papers found would be relevant for the study, the abstract and the conclusion were first studied to see if they contained the answer to the pre-defined questions. For validity and reliability purposes, the entire content of the paper was next scanned to see if it encompassed at least one of the following terms in each subset:

- 1- Sustainability, resilience, long-term, or long-run;
- 2- Environmental, energy, carbon, gas emission, pollution, fossil, fuel, solid waste, green, footprint, or animal welfare;
- 3- Social, security, health, education, housing, equity, population, human rights, worker, staff, labor, safety, or job creation; and
- 4- Collaboration, collaborative, partnership, integrated, alliance, cooperation, or relationship.

Other key terms regarding collaboration and collaborative mechanisms such as information sharing, horizontal versus vertical collaboration, joint planning, etc., were also taken into account.

# 2.4 Results synthesis and report

From the extensive number of papers found, 190 papers answered the requirements. Of these papers, 187 presented specific collaborative mechanisms. Table A1 (Appendix A) presents the 190 papers that address the role of collaboration in improving supply chain sustainability.

Figure 1 demonstrates the number of relevant papers published per year. As this figure shows, the importance of research on collaboration in sustainable supply chains has increased prominently during recent years. The number of papers in 2017 is lower than 2016 as this literature review was completed in January 2017.



Fig. 1. Distribution of reviewed articles based on the year of publication (review completed on 2017-01-25)

Figure 2 illustrates the distribution of papers regarding the country of origin of the first author. As this figure reports, they come from 37 different countries. However, the majority of the studies were done in the USA (33 studies), followed by the United Kingdom (17 studies), and China (13 studies).



Fig. 2. Reviewed articles based on the country of origin of the first author

The next section presents the answers to the first and the second questions of the research based on all the relevant papers found.

# 3. Collaboration to enable sustainability of supply chains

The first and the second questions of the present research, which were to verify the usefulness of collaboration in supply chain sustainability initiatives and the methodology used to support the demonstration, can now be addressed. In this article, we consider a supply chain sustainability initiative to be a long-term arrangement involving at least two partners.

# 3.1 Collaboration in supply chain sustainability initiatives

The systematic literature review that was conducted identified 190 papers recognizing that collaboration had indeed played a role in implementing a supply chain sustainability initiative. For example, Aschemann-Witzel *et al.* (2017) studied the key success factors for supply chains trying to reduce consumer-related food-waste by analyzing 26 existing initiatives. All actors declared that collaboration was of vital importance to the success of their initiatives on reducing consumer food-waste. Reefke and Sundaram (2017) used the Delphi method to survey 15 academics and 20 industrial practitioners on the key enablers of sustainable supply chains and collaboration was found to be a central element. Gimenez *et al.* (2012) analyzed data from a survey of 519 assembly plants from 17 countries and determined from their responses that supply chain collaboration has a statistically significant impact on economic, environmental, and social performance.

Furthermore the different articles investigated highlighted the fact that the collaboration may take multiple inter-organizational forms depending on the sustainable goals pursued by the actors. Some sustainability initiatives involve current supply chain partners, such as reducing greenhouse gas emissions by coordinating with suppliers for optimized deliveries, and may therefore be referred to as **vertical collaboration** (Amer and Eltawil, 2014; Gavronski *et al.*, 2011; and 6 other papers). Other sustainability initiatives rather involve partners who are not within the supply chain, such as reducing greenhouse gas emissions by integrating collaborative consolidation centers shared with other firms, which is referred to as **horizontal collaboration** (Patala *et al.*, 2014; Solakivi *et al.*, 2013; and 4 other papers).

If the firm decides to implement a sustainability initiative with partners from academia, government, non-governmental organizations (NGOs), third-party logistics (3PLs) or other firms in the search for expertise outside of its field, authors in the literature refer to this type of collaboration as **cross-sectoral** (Pehlken *et al.*, 2016; Sahamie *et al.*, 2013; and 6 other papers). For example, Pehlken *et al.* (2016) point out that to implement their sustainable bioenergy solution, one of the main elements is to identify the right actors and to build a strong collaboration between them (e.g. a farmer to provide biomass, a biogas producer, and a bioenergy consumer).

Other types of collaboration have also been mentioned in the literature on sustainable supply chain collaboration, such as **global network collaboration** and **cross-border collaboration** for initiatives involving partners from other countries (Seuring and Gold, 2013; Zander *et al.*, 2016; and 2 others). Tsoi (2010) mentions that language may be an issue to address in these collaborative partnerships. Some also mention **cross-functional collaboration**, especially intra-firm, for initiatives involving employees with different capabilities (Laarie *et al.*, 2016; Ramanathan *et al.*, 2014; and 2 others). Similarly, **inter-disciplinary collaboration** refers to the need for each partner in the initiative to use their specific expertise (Kulak *et al.*, 2016; Sahamie *et al.*, 2013).

We therefore conclude that collaboration is indeed useful to improve the sustainability of a supply chain, that it has been recognized as such by academics as by the industry, and that it may take multiple inter-organizational forms depending on the context considered.

# 3.2 Analysis of the methodologies found in this SLR

Concerning the methodology followed by the authors to demonstrate the collaboration usefulness in sustainable supply chains, we were able to classify them into five groups: case study, mathematical and simulation modeling, qualitative methodology, quantitative methodology, and structured literature review. The case study group is dedicated to empirical research that targeted a phenomena within an industry or a company. Mathematical and simulation modeling considers research that focused on modeling the problem via mathematical modeling, optimization or simulation framework. Quantitative methodology introduces articles that applied statistical analysis, footprint calculation, conceptual frameworks, etc. The qualitative methodology group is allocated to research that explored the reasons and motivations to answer specific problems based on interviews, observations, discussions etc. The structured literature review addresses articles reviewing previous research and their results to answer predefined research questions and highlight future research avenues. Figure 3 presents the distribution of papers according to the applied methodologies of the authors' research and the year of the publication.



Fig. 3. Distribution of reviewed articles based on the methodology and the year of publication

As we can see in this figure, prior to 2015, many studies had applied a case study approach, confirming the interest for companies to find a practical way to achieve higher sustainability goals and the interest for academics to see how they were doing it. However, since 2015, most of the studies have been using qualitative or quantitative methodologies so the distribution in these three methodologies is now more uniform. We also note an increase in contributions using a mathematical or simulation modeling methodology in recent years.

#### 4. Collaborative mechanisms

The third question to answer with this research concerned the collaborative mechanisms that supply chain members could adopt to improve their supply chain sustainability. Regarding this issue, 187 of the 190 papers proposed the use of at least one collaborative mechanism. Based on the variety of mechanisms proposed by the authors, it can be concluded that companies can collaborate through a vast spectrum of methods. In all, a total of 715 collaborative mechanisms were mentioned. Although many seemed to be recurring, at first glance it was difficult to see through this cloud of mechanisms to identify patterns or trends. Therefore, our first step was to sift through all of the collaborative mechanisms to attempt a categorization. Some more general collaborative mechanisms such as joint practices or information sharing were mentioned in several contributions. Other authors preferred to target more specific mechanisms which seemed to be specific instances of joint practices. Likewise, integration of IT solutions and exchanging transactional information seemed to be part of information sharing. The classification exercise revealed certain trends.

Hence, the vast spectrum of collaborative mechanisms found in this SLR was classified into these seven collaborative mechanism categories:

- i) Relationship management (RM);
- ii) Contractual and economic practices (CEP);
- iii) Joint practices (JP);
- iv) Technological and information sharing practices (TISP);

- v) Governance practices (GP);
- vi) Assessment practices (AP); and
- vii)Supply chain design (SCD).

Each category is defined below with examples of specific mechanisms and their roles in sustainable supply chains.

### 4.1 Relationship management

Inspired by Forkmann *et al.* (2016), the collaborative mechanisms in this block are dedicated to activities, organizational processes or routines that facilitate the management of relationships among stakeholders. These mechanisms help raise awareness for sustainability and maintain or increase the support from top management, employees, suppliers, customers, and other stakeholders to address potential issues.

Relationship management seems to be key to successful collaboration since at least one collaborative mechanism in this category is mentioned by almost 80% of the contributions in the SLR. Of these, 30 contributions simply mention the importance of maintaining a good relationship with the other parties involved in the initiative (Koh *et al.*, 2012; Laari *et al.*, 2016; and 28 others).

Communication is one of the main mechanisms in the RM category as it is seen as the foundation of successful collaborative partnerships (Acquaye *et al.*, 2015; Adams, 2008; and 65 others). Communication is also needed to raise awareness, whether it be for suppliers, customers, employees or other stakeholders (Davidson *et al.*, 2014; Gallear *et al.*, 2015; and 12 others). Specific mechanisms used to communicate in a formal or informal manner are meetings, exchange venues, and networking (Bisogno, 2016; Dangelico *et al.*, 2013; and 18 others) whereas more structured mechanisms such as training sessions, workshops, seminars, and education may also be used (Cao *et al.*, 2017; Dai *et al.*, 2015; and 18 others).

The second main mechanism in the RM category is integration (Matt *et al.*, 2014; Molina-Besch and Pålsson, 2016; and 55 others). Interaction, which seems to require less effort than integration, is a collaborative mechanism mentioned by a few authors (Egels-Zandén *et al.*, 2015; Hazelton *et al.*, 2013; and 3 others). Increasing the interaction between collaborating parties could indeed be an effective way of evolving from a collaborative partnership with efficient two-way communication to a completely integrated, smooth-running operation. Integration seems to lead to a deeper collaboration involving other mechanisms such as learning from other companies (Dale *et al.*, 2016; Fiorino and Bhan, 2014; and 7 others), acquiring their knowledge and skills, or developing new ones together (Aschemann-Witzel *et al.*, 2017; Rizzi *et al.*, 2013; and 2 others). According to Gunasekaran *et al.* (2015), green integration is achieved when collaborators align their sustainability goals and create a synergy through synchronized actions.

Another key collaborative mechanism for the success of a sustainability initiative is motivation (Airike et al, 2016; Hall *et al.*, 2013; and 6 others). Collaborative mechanisms that will nourish this motivation are the firms' commitment to the partnership (Leadbitter and Benguerel, 2014; Pimenta and Ball, 2015; and 2 others) and partnership support (Aguiar *et al.*, 2013; Maloni and Benton, 1997; Pimenta and Ball, 2015) such as sponsoring meetings for suppliers to share information (Sancha *et al.*, 2016) or investing in formal relationship building activities (Touboulic and Walker, 2015). Several contributions mentioned that the relationship should be built on a mutual understanding (Bernstein and Cashore, 2007; Blome *et al.*, 2014; and 14 others) and shared values (Reefke and Sundaram, 2017).

Finding the right collaboration partner is also vital to a sustainable collaborative relationship. When a firm must find the right partners to implement a new sustainability initiative, several authors identified joining collaboration networks or associations as an efficient collaborative mechanism (Dangelico *et al.*, 2013; Fischer and Pascucci, 2017; and 7 others). As a specific example, O'Keefe *et al.* (2016) describe how the Great Deal network, designed to encourage energy efficiency measures in the United Kingdom, is used to bring potential business partners together. A few contributions describe the relationship characteristics that could lead to good collaboration, such as mutual trust (Dan *et al.*, 2010; Li and Found, 2016; and 5 others), goodwill and openness (Gunasekaran *et al.*, 2015; Maloni and Benton, 1997; Wan Ahmad *et al.*, 2016), transparency and proactive behaviour (Wan Ahmad *et al.*, 2016), flexibility and solidarity (Dan *et al.*, 2010), organizational compatibility (Youn *et al.*, 2013) as well as a mentality to embrace change (Reefke and Sundaram, 2017).

# 4.2 Contractual and economic practices

Contracts can be conceptualized as any written or spoken and legal agreement among stakeholders. A contract forces all parties to accept their responsibilities and respect their promises (e.g. a costprofit sharing contract between a manufacturer who must invest in the design of an eco-product and a retailer who will profit from selling it). Economic practices such as incentive alignments, rewards, and cost-profit sharing are also included in this category. The mechanisms in this category will help motivate long-term and steady collaboration between companies as well as clearly indicate expectations from each party.

Establishing a legal contract is a collaborative mechanism mentioned in 95 contributions which represents 50% of the selected papers. Most simply refer to contracts in general (Shanoyan *et al.*, 2014; Tarandi, 2015; and 36 others). However, some refer to agreements (Mengistie *et al.*, 2015; Nwaka, 2005; and 33 others), revenue sharing contracts (Jakhar, 2015; Klassen and Vachon, 2003; and 3 others), cost sharing contracts (Ghosh and Shah, 2015; Tesfay, 2014; Yenipazarli, 2017), long-term contracts (Bai *et al.*, 2012; Cheung *et al.*, 2009; Davidson *et al.*, 2014), multilateral contracts (Brown *et al.*, 2007), and joint ventures (Roy and Whelan, 1992; Umar *et al.*, 2013; and 5 others).

Economic practices are also collaborative mechanisms frequently mentioned in this category, some form or other being mentioned in 91 contributions. Economic incentives, a more general term, is by far the most frequently mentioned (Sime, 2005; Sippl, 2015; and 56 others). Other, more specific types found in the literature are sharing benefits (Koh *et al.*, 2012; Maloni and Benton, 1997; and 12 others), cost-profit sharing (Amer and Eltawil, 2014; Basiri and Heydari, 2017; and 2 others), incentive alignment (Brockhaus *et al.*, 2013; Nematollahi *et al.*, 2017b; and 2 others), and rewards (Formentini and Taticchi, 2016; Hussain *et al.*, 2015; Maloni and Benton, 1997).

For example, Ding *et al.* (2015) studied the pricing strategy of sustainable supply chains and explored a mechanism to motivate supply chain actors to produce environmentally green products. Via a mathematical modeling methodology, and by applying regulation and policy incentives, they showed that cooperation of supply chain actors enhance environmental performance through collaborative adjustment of transfer price considering government policy incentive.

# 4.3 Joint practices

In this research, joint practices address common, bilateral, and mutual activities among the parties involved. The spectrum of cooperation practices ranges from joint planning up to joint control activities (Mentzer, 2001). The mechanisms in this category facilitate collaboration because they rely

on the knowledge and the expectations of all parties involved to establish a consensus to which all parties will adhere to, be it a common production plan, a common forecast, etc.

Of the contributions analyzed in this SLR, 122 papers mentioned a collaborative mechanism in the JP category. Some authors referred to it as joint-effort, joint-execution, and cooperation (Brockhaus *et al.*, 2013; Burlingame and Pineiro, 2007; and 42 others). Some others mentioned the importance of having all parties participate in planning the initiative (Albino *et al.*, 2012; Fischer *et al.*, 2016; and 17 others), joint solution finding (Airike *et al.*, 2016; Gold *et al.*, 2010; and 10 others) as well as collaborative decision-making (Basiri and Heydari, 2017; Chan *et al.*, 2012; and 15 others). Several authors also specify that for the collaboration to be successful, the parties involved must share a common goal that is clear to all (Cheung *et al.*, 2009; Dale *et al.*, 2016; and 24 others).

This category also includes the concept of sharing. Many authors mention that sharing activities (van Hoof and Thiell, 2014; Vrijhoef *et al.*, 2014; and 3 others), sharing resources (Hajdul, 2014; Nathanail *et al.*, 2016; and 4 others), and sharing processes (Akhavan and Beckmann, 2017; Chan *et al.*, 2012) can enhance collaboration between the actors of a sustainable initiative. We also consider that developing mutually beneficial capabilities, found in 11 contributions such as Gallear *et al.* (2015) and Kovács (2008), is a collaborative mechanism in the JP category.

A few authors target specific sustainability initiatives and are then able to propose more precise JP mechanisms. For example, two firms could collaborate on an eco-product development initiative (Gunasekaran et al., 2015; Kulak et al., 2016; and 12 others). Kulak et al. (2016) takes this concept further and proposes mechanisms such as integrative design and a collaborative design workshop. Similarly, a firm could collaborate with suppliers to ensure eco-sourcing (Teixeira et al., 2016; Yan et al., 2016; and 3 others), help the supplier develop to meet sustainability standards (Akhavan and Beckmann, 2017; Wan Ahmad et al., 2016) or even manage delivery schedules and routes together (Siddiqui and Raza, 2015; Tseng and Bui, 2017). In the hopes of reducing inventories, warehouse sizes, energy consumption, greenhouse gas emissions and/or waste in general, supply chains may aim for more sustainable operations management. Some of the JP mechanisms proposed in this case are collaborative forward flow of products (Amaro and Barbosa-Póvoa; 2013; Gunasekaran et al., 2015; Ytterhus et al., 1999) which might benefit from joint forecasting or joint production planning (Chkanikova, 2015; Jiang et al., 2016; and 4 others) and synchronization (Musa et al., 2014; Reefke and Sundaram, 2017; and 3 others). A specific example of joint planning is given by Réviron and Chappuis (2005) where a popular food chain in Switzerland has partnered with a meat producer association in order to guarantee quality of meat and reduce food loss. The association and the food chain executives meet regularly to discuss the quantity of meat to produce. Furthermore, both agree on a monthly quantity that the food chain accepts to purchase from the meat producers.

# 4.4 Technological and information sharing practices

This category is based on the definition of Montoya-Torres and Ortiz-Vargas (2014) who describe collaboration and information sharing in supply chains and defined technological and information sharing practices as the activities, data, and tools used in facilitating the procedure of collaboration among different entities of a supply chain. The collaborative mechanisms in this category aim to facilitate the collaboration in sustainability initiatives by providing relevant and accurate information when it is needed. The relevance, accuracy, and timeliness of the information shared will depend on the collaborative mechanisms that are put in place.

Information sharing is the most frequently mentioned collaborative mechanism in this category. A total of 69 contributions emphasized its importance (Acquaye *et al.*, 2015; Barari *et al.*, 2012; and

67 others). Some authors also indicate knowledge sharing (Fischer and Pascucci, 2017; Gallear *et al.*, 2015; and 14 others) to deepen collaboration. Whereas information sharing could refer to simple data exchange, knowledge sharing implies transferring or teaching abilities and know-how that would require closer collaboration.

As for technological collaborative mechanisms, several authors address the need for an information system or a platform allowing for easy exchange of data (Brown *et al.*, 2007; Chenga, 2011; and 39 others). These two contributions, along with 7 others, specify that if there are many systems, they should be integrated with seamless communication between them. Among the technological collaborative mechanisms mentioned by contributions in this SLR, the most frequent are a standardized model for visualizing and sharing information (Hu *et al.*, 2015, Klassen and Vereecke, 2012; Kurdve *et al.*, 2015; and 3 others), data harmonization and analytics (De Souza *et al.*, 2014; Fischer *et al.*, 2016; and 2 others), ensuring the quality of the information exchanged (Le Dû and de Corbière, 2011; Li and Found, 2016; Waller *et al.*, 2015) as well as providing technical assistance to all parties involved (Dai *et al.*, 2015; Hall *et al.*, 2013; Sigala, 2008).

Once again, studies focusing on specific collaborative sustainability initiatives mention the use of more precise technological collaborative mechanisms such as a product information system (Boström *et al.*, 2015; Waller *et al.*, 2015), an intelligent transportation system (Bucklew, 2011), an e-commerce platform (Goyal *et al.*, 2017; Jiang *et al.*, 2016; Musa *et al.*, 2014) or a website-forum (Kumar and Malegeant, 2006; Hall *et al.*, 2013). For example, Schniederjans and Hales (2016) mention cloud-computing as a collaborative mechanism. They surveyed 247 IT and supply chain professionals and concluded that cloud computing can indeed be used to positively impact collaboration, and is positively associated with both economic and environmental performance.

## 4.5 *Governance practices*

The collaborative mechanisms related to governance practices address rules, trends, policies, and administrative laws that control, direct or manage the activities, organizations, and systems. Decision makers of private sectors also might define a set of policies, guidelines, and standards, with respect to public policies, to manage and make decisions regarding the contexts, events, and interactions among stakeholders. These collaborative mechanisms facilitate collaboration in sustainability initiatives as they help specify the common goals and the responsibilities of each party. Establishing policies and standards may also help in having a higher level of process consistency, innovation, responsiveness and transparency, better costs and benefits alignment, and more efficient environmental monitoring.

Of all the contributions in this SLR, 105 mentioned a collaborative mechanism from this category. These mechanisms can be separated into two main groups: i) governance from external entities such as the government or environmental organisations and ii) governance from within the partnership such as internal sustainability standards and practices.

To begin with, several authors mentioned governance as an efficient collaborative mechanism without necessarily indicating whether it was external governance, internal governance, or both (Bernstein and Cashore, 2007; Gold *et al.* 2010; and 16 others). However, Boström *et al.* (2015) do specify that they consider governance to be guidelines, rules, norms, standards, and exercising authority while sustainable governance refers to eco-labels, codes of conduct, procurement guidelines, and eco-branding. The external collaborative mechanisms that we found are government policies, regulations and legislation (Halloran *et al.*, 2014; Iacob, 2015; and 8 others), government subsidies (Ellram and Golicic, 2016; Forsman *et al.*, 2014; and 4 others), government policy incentives (Ding *et al.*, 2016a; O'Keefe *et al.*, 2016; Yan *et al.*, 2016), government sponsored training (Cao *et al.*, 2017; Tseng and Bui, 2017), NGO supplier development programs (Rodríguez *et al.*, 2016), adhesion to international initiatives such as Global Compact (Formentini and Taticchi, 2016), and any other help received from the government, agencies, and the community to support the sustainability initiative such as funding, data or information (Tseng and Bui, 2017).

As an illustration, Korhonen *et al.* (2015) studied the role of environmental regulations in the future competitiveness of the pulp and paper industry through a qualitative methodology. The regulatory measures were perceived as environmentally advantageous and effective in the long term. The role of the policy implementation in achieving actual environmental improvement was also emphasized.

The most frequent internal governance collaborative mechanism is establishing standards (Adams, 2008; Brown *et al.*, 2007; and 54 others) to set the performance levels that are expected from each party. Other recurrent collaborative mechanisms are putting into place an effective governance structure (Barari *et al.*, 2012; Gimenez *et al.*, 2012; and 17 others) and standardizing the methods and processes (Egels-Zandén *et al.*, 2015; Kerr and Foster, 2011; and 5 others).

### 4.6 Assessment practices

Based on a study by Sancha *et al.* (2016) on assessing specific supplier performance, the collaborative mechanisms in this block concern activities that enable collaboration for supply chain sustainability via evaluation and monitoring. However, selecting a particular partner because of its sustainably-oriented practices would rather be considered in the supply chain design mechanism category, as explained in the following subsection. Collaborative mechanisms in the AP category help firms evaluate their performance and the performance of their partners so that they can make adjustments when needed.

Collaborative mechanisms in the AP category were found in only 7% of the contributions in this SLR. The most frequently mentioned collaborative mechanism in this category is sustainability performance assessment (Fischer et al., 2016; Kurdve et al., 2015; and 17 others). For example, a particular sustainability initiative might require life-cycle assessment (Bonou et al., 2016; Formentini and Taticchi, 2016; and 2 others) or maybe green supplier assessment (Li and Found, 2016; Siddiqui and Raza, 2015; Tesfay, 2014). According to the sustainable performance indicators that are required, monitoring procedures (Chopra and Wu, 2016; Blanco et al., 2016; and 14 others) and auditing procedures (Akhavan and Beckmann, 2017; Maloni and Benton, 1997; and 3 others) are collaborative mechanisms that could be put in place. Environmental certification is also another collaborative mechanism that can be used either as performance expectations in a particular initiative or as selection criteria for potential sustainability initiative partnerships (Gallear et al., 2015; Gavronski et al., 2011; and 11 others). One of the benefits of performance assessment is that partners develop a mutual understanding of accountability for environmental performance (De Giovanni and Vinzi, 2012 and 2014; Tseng and Bui, 2017). The performance results can then be used to prioritize process optimizations (De Giovani and Vinzi, 2014; Pimenta and Ball, 2015; and 2 others). To this effect, a few contributions proposed optimization models for sustainability initiatives such as Soysal's (2016) closed-loop inventory routing optimization model that considers fuel consumption or Türkay et al.'s (2004) energy sharing optimization model.

# 4.7 Supply chain design

Sustainable supply chain design will take both environmental (gas emissions, waste reduction, etc.) and social (job creation, social repercussions, etc.) aspects into consideration when being established (e.g. choosing a building location, closing a factory, choosing a supplier, etc.). The collaborative mechanisms in this category will facilitate collaboration in supply chain sustainability initiatives by offering decision-making tools to supply chain partners. In turn, a supply chain designed by taking into account sustainability issues will offer a more collaborative environment for present and future sustainability initiatives.

The most frequently mentioned collaborative mechanism in this category is selecting sustainabilityoriented partners (Amer and Eltawil, 2014; Goyal *et al.*, 2017; and 4 others) or more specifically sustainability-oriented suppliers (Aguiar *et al.*, 2013; Bonou *et al.*, 2016; and 16 others). For example, Govindan *et al.* (2013) proposed a model to measure the sustainability performance of a supplier based on economic, environmental, and social aspects to help with supplier selection operations. Tseng and Bui (2017) also mentioned that geographic proximity can be considered when choosing a supplier as it favors trust and collaboration while minimizing costs and impacts on material flow.

A few contributions addressed supply chain design directly, stating that supply chain design or facility location decisions should be taken in collaboration with other supply chain partners (Amer and Eltawil, 2014; Ding *et al.*, 2016a). Choosing a transportation mode could also consider sustainability issues (Siddiqui and Raza, 2015) or a firm could decide to partner with an eco-certified transportation provider (Ellram and Golicic, 2016).

Collaborative mechanism	Description	Examples of collaborative mechanisms	No. of studies
Relationship management	Activities that facilitate the management of relationships among stakeholders	Communication, raising awareness, meet- ings, networking, training, integration, motivation	150
Contractual and economic prac- tices	Written or spoken agreement among stakeholders with or without incentives	Contracts, incentive alignments, rewards, cost-profit sharing	127
Joint practices	Common, bilateral, and mutual activities among stakeholders	Joint effort, joint planning, joint solution finding, collaborative decision-making, common goal, sharing resources	122
Technological and information sharing practices	Activities regarding the data and tools used in facilitating the procedure of col- laboration among different entities of a supply chain	Information and knowledge sharing, in- formation systems, platforms	113
Governance prac- tices	Rules, trends, policies, and administra- tive laws to control, direct or manage the activities, organizations, and systems	Laws, policies, guidelines, regulations, performance standards, governance struc- ture	105
Assessment prac- tices	Activities that increase the quality of performance in supply chains via evalua- tion and monitoring	Performance assessment, life cycle as- sessment, supplier assessment, monitoring, audit, certification	54
Supply chain design	Considers an environmentally and social friendly structure for the supply chain	Supplier selection, facility location deci- sions	44

Table 1 demonstrates the distribution of the studies regarding the proposed seven categories.

# 5. Discussion

The articles found in this research confirmed the key role collaboration may play in sustainability initiatives. One of the particularities of interfirm collaborations for sustainable supply chains is that they usually involve partners from outside the supply chain and many stakeholders, in a context where considering environmental and social aspects is usually less profitable in the short term (Albino *et al.*, 2012). According to many authors, addressing complex sustainability issues requires network-based collaboration based on a common goal and sustainability performance standards. Putting in place monitoring procedures and contracts to enforce these standards also seem to be of utmost importance for sustainability initiatives.

If we look at the mechanisms more specifically, among the 190 papers recognizing the role of collaboration in moving a current supply chain toward a more sustainable one, 187 papers addressed at least one collaborative mechanism. In fact, a detailed analysis of how many collaborative mechanism categories were addressed by each contribution demonstrated that the majority of contributions mentioned collaborative mechanisms from at least 4 categories. This seems to indicate that combining mechanisms from different categories would be more beneficial than targeting mechanisms from only one category. Figure 4 illustrates the percentage of contributions mentioning collaborative mechanisms from a given number of categories.



Fig. 4. Percentage of contributions mentioning collaborative mechanisms from a given number of categories

A good example is the work of Luzzini *et al.* (2015) in which the role of intra- and interfirm collaborative capabilities in the upstream supply chain was studied through a set of reflective indicators such as cross-functionality of decision making for supply market analysis (JP category), sourcing strategy (GP category), supplier selection (SCD category), contracting (CEP category), and supplier evaluation (AP category). Via a quantitative methodology and the analysis of data from 383 procurement executives of European and North American countries, they showed the links between sustainability commitment (RM category) and both intra- and interfirm collaborative capabilities as well as between interfirm collaborative capabilities and environmental, social, and cost performance.

The set of mechanisms that each contribution mentioned seemed to depend on the sustainability initiative's objective, the business environment complexity considered as well as the level of trust and effort stakeholders are willing to put forward. Tesfay (2014) for instance studied the strategy of sea transport outsourcing and in particular how an oil and gas company coordinates with such a strategy. Via the case study, they showed that the best solution regarding environmentally friendly sea transport service is to select certified suppliers, make partnerships, and provide contracts with cost refund agreements. Lind et al. (2016) addressed a much more complex sea transport sustainability initiative. They proposed an infrastructure that would allow standardized real-time information sharing between ships and ports to support the Sea Traffic Management (STM) sustainability initiative. STM aims to share relevant and timely maritime information between service providers so that they may be aware of maritime situations, improve predictability of arrivals and departures, plan just-in-time operations, and provide innovation capability. The collaborative mechanisms found in this article cover all 7 categories: providing situational awareness, benefit sharing, collaborative decision-making, information sharing, holistic sea traffic management approach, international maritime standards, continuously monitoring and adjusting voyage plan, traffic coordination, and port call synchronization/optimization.

While all the mechanisms found seem to play a role in improving supply chain sustainability, many authors have also pointed out non-negligible challenges to face, such as the lack of trust between supply chain members, the level of commitment and the lack of interest to share risks, the need for training and visibility, a high level of required bureaucracy, the presence of a traditional management philosophy, incompatible technology, project-based context of work (short-term tasks), and unfair benefit sharing (see for example Walker *et al.*, 2008 or O'Keefe *et al.*, 2016). As a result, even though the results highlighted in the literature show the positive impact collaboration may have on sustainability initiatives, businesses should still keep in mind these challenges when selecting and deploying their collaborative mechanisms.

# 6. Conclusion

In this research, a systematic literature review was conducted to identify the papers that addressed the role of collaboration in improving supply chain sustainability. Research questions and criteria were determined and a set of keywords and pre-defined search engines selected. Based on search results, an extensive number of papers were evaluated and 190 of them were identified as relevant according to the research questions. All accepted papers were analyzed and classified based on the year of publication, the country of origin of the first author and the methodology used. Among all selected papers, 187 of them addressed specific collaborative mechanisms. These mechanisms were classified into seven categories: relationship management, contracts and economic practices, joint practices, technology and information sharing practices, governance practices, assessment practices, and supply chain design.

Results from the systematic literature review confirmed that collaboration is useful to improve the sustainability of a supply chain, that it has been recognized as such by academics as by the industry, and that it may take multiple inter-organizational forms depending on the context considered. An analysis of the methodologies used in the contributions revealed that most of the authors used case studies. However, in recent years, we notice a definite increase in the number of contributions with qualitative, quantitative and mathematical or simulation modeling methodologies. Regarding the mechanisms, the top five most frequently mentioned, in decreasing order, are contracts, economic incentives, communication, information sharing, and integration. Another striking result is that 96%

of the contributions mentioned collaborative mechanisms from more than one category. This seems to indicate that combining mechanisms from different categories would be more beneficial than targeting mechanisms from only one category.

Results of this research may help future practitioners to recognize the effective collaborative mechanisms for improving their supply chain sustainability. They also highlight interesting challenges that could certainly be studied more deeply: the choice of the mechanisms to implement depending on the maturity level of each partner, the periods and the frequency to efficiently combine different mechanisms over time, the most appropriate mechanisms to select regarding the structure of the inter-organizational collaboration considered, the trade-off between the level of investment available and the complexity of the collaborative mechanisms implemented, the development of the right KPIs to track mechanisms from both an individual and a supply chain point of view, the investigation of case studies to evaluate the effectiveness of the proposed collaborative mechanisms, as well as an evaluation of the mechanism interconnections via statistical analysis.

#### References

- Abbasi, B., Farsijani, H., Raad, A., 2016. Investigating the effect of supply chain management on sustainable performance focusing on environmental collaboration. Modern Applied Science 10 (12), 115-126.
- Acquaye, A.A., Yamoah, F.A., Feng, K., 2015. An integrated environmental and fair trade labelling scheme for product supply chains. Int. J. Prod. Econ. 164, 472-483.
- Adams, R.J., 2008. Fast food and animal rights: An examination and assessment of the industry's response to social pressure. Business and Society Review 113 (3), 301-328.
- Aguiar, J., Pimenta, C., Ball, P., 2013. Green Supply Chain Management: A Study on Criteria Selection and Collaboration with Suppliers. International workshop advances in cleaner production, Brazil, May 22-24.
- Airike, P.-E., Rotter, J.P., Mark-Herbert, C., 2016. Corporate motives for multi-stakeholder collaboration corporate social responsibility in the electronics supply chains. J. Clean. Prod. 131, 639-648.
- Akhavan, R.M., Beckmann, M., 2017. A configuration of sustainable sourcing and supply management strategies. J. Purchasing & Supply Management, In Press, http://dx.doi.org/10.1016/j.pursup.2016.07.006.
- Albino, V., Dangelico, R.M., Pontrandolfo, P., 2012. Do inter-organizational collaborations enhance a firm's environmental performance? a study of the largest US companies. J. Clean. Prod. 37, 304-315.
- Amaro A.C.S., Barbosa-Póvoa, A.P.F.D., 2013. Supply Chain Management Optimal Planning for Sustainable Products' Portfolio. Computer Aided Chemical Engineering 32, 559-564.
- Amer, L.E., Eltawil, A.B., 2014. Collaborative sustainable supply chain network design: state of the art and solution framework. Proceedings of CIE44 & IMSS'14.
- Anthony, V.M., Ferroni, M., 2012. Agricultural biotechnology and smallholder farmers in developing countries. Current opinion in biotechnology 23 (2), 278-285.
- Aschemann-Witzel, J., de Hooge, I.E., Rohm, H., Normann, A., Bossle, M.B., Grønhøj, A., Oostindjer, M., 2017. Key characteristics and success factors of supply chain initiatives tackling consumer related food waste – A multiple case study. J. Clean. Prod., In Press, doi: 10.1016/j.jclepro.2016.11.173.
- Audy, J.-F., Lehoux, N., D'Amours, S., Rönnqvist, M., 2012. A framework for an efficient implementation of logistics collaborations. International Transactions in Operational Research, 19, 633-657.
- Bai, Y., Ouyang, Y., Pang, J.S., 2012. Biofuel supply chain design under competitive agricultural land use and feedstock market equilibrium. Energy Economics 34 (5), 1623-1633.
- Barari, S., Agarwal, G., Zhang, W.C., Mahanty, B., Tiwari, M.K., 2012. A decision framework for the analysis of green supply chain contracts: An evolutionary game approach. Expert systems with applications 39 (3), 2965-2976.
- Barnes, S.J., Mattsson, J., 2016. Understanding current and future issues in collaborative consumption: A four-stage Delphi study. Technological Forecasting and Social Change 104, 200-211.
- Barratt, M. 2004. Understanding the Meaning of Collaboration in the Supply Chain. Supply Chain Management: An International Journal 9 (1), 30–42.
- Basiri, Z., Heydari, J., 2017. A mathematical model for green supply chain coordination with substitutable products, J. Clean. Prod., In Press, doi: 10.1016/j.jclepro.2017.01.060.
- Bernstein, S., Cashore, B., 2007. Can non-state global governance be legitimate? An analytical framework. Regulation & Governance 1 (4), 347-371.
- Bishop, A., Riopelle, K., 2011. Identifying Potential Collaborative Innovation Networks to Support Emerging Chemical Legislation and Global Design for Sustainability. Procedia-Social and Behavioral Sciences 26, 38-45.
- Bisogno, M., 2016. Corporate Social Responsibility and Supply Chains: Contribution to the Sustainability of Well-being. Agriculture and Agricultural Science Procedia 8, 441-448.
- Blanco, C., Caro, F., Corbett, C.J., 2016. The state of supply chain carbon footprinting: analysis of CDP disclosures by US firms. J. Clean. Prod. 135, 1189-1197.

- Blome, C., Paulraj, A., Schuetz, K., 2014. Supply chain collaboration and sustainability: a profile deviation analysis. Int. J. Op. & Prod. Management 34 (5), 639 663.
- Bonou, A., Laurent, A., Olsen, S.I., 2016. Life cycle assessment of onshore and offshore wind energy-from theory to application. Applied Energy 180, 327-337.
- Boström, M., Jönsson, A. M., Lockie, S., Mol, A.P., Oosterveer, P., 2015. Sustainable and responsible supply chain governance: challenges and opportunities. J. Clean. Prod. 107, 1-7.
- Brockhaus, S., Kersten, W., Knemeyer, A.M., 2013. Where do we go from here? Progressing sustainability implementation efforts across supply chains. J. Business Logistics 34 (2), 167-182.
- Brown, J. E., Hendry, C. N., Harborne, P., 2007. An emerging market in fuel cells? Residential combined heat and power in four countries. Energy policy 35 (4), 2173-2186.
- Bucklew, K.J., 2011. Improving Freight Roadway Transportation with Dedicated Truck Lanes: Opportunities and Issues. Transportation Journal 50 (4), 431-445.
- Burlingame, B., Pineiro, M., 2007. The essential balance: Risks and benefits in food safety and quality. Journal of Food Composition and Analysis 20 (3), 139-146.
- Cao, H., Scudder, C., Dickson, M.A., 2017. Sustainability of Apparel Supply Chain in South Africa: Application of the Triple Top Line Model. Clothing and Textiles Research Journal, 1-17, DOI: 10.1177/0887302X16688560.
- Cao, J., Zhang, X., 2013. Coordination strategy of green supply chain under the free market mechanism. Energy Procedia 36, 1130-1137.
- Cao, J., Zhang, X., Zhou G., 2015. Supply chain coordination with revenue-sharing contracts considering carbon emissions and governmental policy making. Environmental Progress & Sustainable Energy, DOI 10.1002/ep | Published online 2015 in Wiley Online Library (wileyonlinelibrary.com). DOI 10.1002/ep.12246.
- Chan, F.T., Chong, A.Y.L., Zhou, L., 2012. An empirical investigation of factors affecting e-collaboration diffusion in SMEs. Int. J. Prod. Econ. 138 (2), 329-344.
- Chenga, J. C., 2011. A web service framework for measuring and monitoring environmental and carbon footprint in construction supply chains. Proceedia Engineering 14, 141-147.
- Cheung, D.K., Welford, R.J., Hills, P.R., 2009. CSR and the environment: business supply chain partnerships in Hong Kong and PRDR, China. Corporate Social Responsibility and Environmental Management 16 (5), 250-263.
- Chin, T.A., Tat, H.H., Sulaiman, Z., 2015. Green Supply Chain Management, Environmental Collaboration and Sustainability Performance. Procedia 26, 695–699.
- Chkanikova, O., 2015. Sustainable Purchasing in Food Retailing: Interorganizational Relationship Management to Green Product Supply. Business Strategy and the Environment 25 (7), 478-494.
- Chopra, S., Wu, P.-J., 2016. Eco-activities and operating performance in the computer and electronics industry. European Journal of Operational Research 248, 971-981.
- Dai, J., Montabon, F.L., Cantor, D.E., 2015. Reprint of "Linking rival and stakeholder pressure to green supply management: Mediating role of top management support". Transportation Research Part E: Logistics and Transportation Review 74, 124-138.
- Dale, V.H., Kline, K.L., Buford, M.A., Volk, T.A., Smith, C.T., Stupak, I., 2016. Incorporating bioenergy into sustainable landscape designs. Renewable and Sustainable Energy Reviews 56, 1158-1171.
- Dan, L., Songzheng, Z., Weiyang, J., 2010. An Empirical Study for Green Supply Chain Management: Relationship between Collaboration and Extended Supply Chain Performance. In Management and Service Science (MASS), 2010 International Conference on (p. 1-4). IEEE.
- Dangelico, R.M., Pontrandolfo, P., Pujari, D., 2013. Developing sustainable new products in the textile and upholstered furniture industries: role of external integrative capabilities. J. Product Innovation Management 30 (4), 642-658.
- Davidson, E.A., Galloway, J.N., Millar, N., Leach, A.M., 2014. N-related greenhouse gases in North America: innovations for a sustainable future. Current Opinion in Environmental Sustainability 9, 1-8.
- De Giovanni, P., Vinzi, V.E., 2012. Covariance versus component-based estimations of performance in green supply chain management. Int. J. Prod. Econ. 135 (2), 907-916.
- De Giovanni, P., Vinzi, V.E., 2014. The benefits of a monitoring strategy for firms subject to the Emissions Trading System. Transportation Research Part D: Transport and Environment 33, 220-233.
- De Sousa Jabbour, A.B.L., Jabbour, C.J.C., Latan, H., Teixeira, A.A., de Oliveira, J.H.C., 2015. Quality management, environmental management maturity, green supply chain practices and green performance of Brazilian companies with ISO 14001 certification: Direct and indirect effects. Transportation Research Part E: Logistics and Transportation Review 74, 139-151.
- De Souza, R., Goh, M., Lau, H.C., Ng, W.S., Tan, P.S., 2014. Collaborative urban logistics-synchronizing the last mile a Singapore research perspective. Procedia-Social and Behavioral Sciences 125, 422-431.
- Delmotte, S., Barbier, J.-M., Mouret, J.-C., Le Page, C., Wery, J., Chauvelon, P., Sandoz, A., Ridaura, S.L., 2016. Participatory integrated assessment of scenarios for organic farming at different scales in Camargue, France. Agricultural Systems, 143, 147-158.
- Diabat, A., Govindan, K., 2011. An analysis of the drivers affecting the implementation of green supply chain management. Resources, Conservation and Recycling 55, 659-667.
- Ding, H., Zhao, Q., An, Z., Xu, J., Liu, Q., 2015. Pricing strategy of environmental sustainable supply chain with internalizing externalities. Int. J. Prod. Econ. 170, 563-575.
- Ding, H., Zhao, Q. An, Z., Tang, O., 2016a. Collaborative mechanism of a sustainable supply chain with environmental constraints and carbon caps. Int. J. Prod. Econ. 181, 191-207.
- Ding, H., Liu, Q., Zheng, L., 2016b. Assessing the economic performance of an environmental sustainable supply chain in reducing environmental externalities. European J. Op. Res., 255, p. 463-480.
- Dissanayake, G., Sinha, P., 2015. An examination of the product development process for fashion remanufacturing. Resources, Conservation and Recycling 104, 94-102.

- Egels-Zandén, N., Hulthén, K., Wulff, G. 2015. Trade-offs in supply chain transparency: The case of Nudie Jeans Co. J. Clean. Prod. 107, 95-104.
- Ellram, L.M., Golicic, S.L., 2016. The role of legitimacy in pursuing environmentally responsible transportation practices. J. Clean. Prod. 139, 597-611.
- Fiorino, D.J., Bhan, M., 2014. Supply Chain Management as Private Sector Regulation: What does it Mean for Business Strategy and Public Policy? Business Strategy and the Environment 25 (5), 310-322.
- Fischer, J.-H., Thomé, A.M.T., Scavarda, L.F., Hellingrath, B., Martins, R., 2016. Development and Application of a Maturity Measurement Framework for Supply Chain Flexibility. Procedia CIRP 41, 514-519.
- Fischer, A., Pascucci, S., 2017. Institutional incentives in circular economy transition: The case of material use in the Dutch textile industry. J. Clean. Prod., in Press, http://dx.doi.org/10.1016/j.jclepro.2016.12.038.
- Flint, D. J., Signori, P., 2014. Environmental sustainability and the supply chain: Is collaboration necessary? Consulted on 2015-03-05 at: <u>http://www.univr.it/documenti/AllegatiOA/allegatooa\_42411.pdf</u>
- Forkmann, S., Henneberg, S.C., Naudé, P., Mitrega, M., 2016. Supplier relationship management capability: a qualification and extension. Industrial Marketing Management 57, 185-200.
- Formentini, M., Taticchi, P., 2016. Corporate sustainability approaches and governance mechanisms in sustainable supply chain management. J. Clean. Prod. 112, 1920-1933.
- Forsman, C., Milani, P., Schondebare, J.A., Matthias, D., Guyondet, C., 2014. Rice fortification: a comparative analysis in mandated settings. Annals of the New York Academy of Sciences 1324 (1), 67-81.
- Gallear, D., Ghobadian, A., He, Q., 2015. The mediating effect of environmental and ethical behaviour on supply chain partnership decisions and management appreciation of supplier partnership risks. Int. J. Prod. Res. 53 (21), 6455-6472.
- Gavronski, I., Klassen, R.D., Vachon, S., do Nascimento, L.F.M., 2011. A resource-based view of green supply management. Transportation Research Part E: Logistics and Transportation Review 47 (6), 872-885.
- Germani, M., Mandolini, M., Marconi, M., Marilungo, E., Papetti, A., 2015. A System to Increase the Sustainability and Traceability of Supply Chains. Procedia CIRP 29, 227-232.
- Ghosh, D., Shah, J., 2015. Supply chain analysis under green sensitive consumer demand and cost sharing contract. Int. J. Prod. Econ. 164, 319-329.
- Gimenez, C., Sierra, V., Rodon, J., 2012. Sustainable operations: Their impact on the triple bottom line. Int. J. Prod. Econ. 140 (1), 149-159.
- Gold, S., Seuring, S., Beske, P., 2010. Sustainable supply chain management and inter-organizational resources: a literature review. Corporate Social Responsibility and Environmental Management 17 (4), 230-245.
- Govindan, K., Khodaverdi, R., Jafarian, A., 2013. A fuzzy multi criteria approach for measuring sustainability performance of a supplier based on triple bottom line approach. J. Clean. Prod. 47, 345-354.
- Goyal, S., Esposito, M., Kapoor, A., 2017. Circular economy business models in developing economies: Lessons from India on reduce, recycle and reuse paradigms. Thunderbird International Business Review, In Press, DOI: 10.1002/tie.21883.
- Grekova, K., Calantone, R.J., Bremmers, H.J., Trienekens, J.H., Omta, S.W.F., 2016. How environmental collaboration with suppliers and customers influences firm performance: evidence from Dutch food and beverage processors. J. Clean. Prod. 112, 1861-1871.
- Grimm, J.H., Hofstetter, J.S., Sarkis, J., 2016. Exploring sub-suppliers' compliance with corporate sustainability standards. J. Clean. Prod. 112, 1971-1984.
- Gunasekaran, A., Subramanian, N., Rahman, S., 2015. Green supply chain collaboration and incentives: Current trends and future directions. Transportation Research Part E: Logistics and Transportation Review 74, 1-10.
- Hajdul, M., 2014. Virtual Co-opetition in Transport-T-Scale Platform Case Study. Procedia-Social and Behavioral Sciences 111, 761-769.
- Hall, P.V., O'Brien, T., Woudsma, C., 2013. Environmental innovation and the role of stakeholder collaboration in West Coast port gateways. Research in Transportation Economics 42 (1), 87-96.
- Halloran, A., Clement, J., Kornum, N., Bucatariu, C., Magid, J., 2014. Addressing food waste reduction in Denmark. Food Policy 49, 294-301.
- Hazelton, J.A., Windhorst, K., Amezaga, J.M., 2013. Forest based biomass for energy in Uganda: Stakeholder dynamics in feedstock production. Biomass and Bioenergy 59, 100-115.
- Hisjam, M., Guritno, A.D., Supriyatno, N., Tandjung, S.D., 2015. A Sustainable Partnership Model among Supply Chain Players in Wooden Furniture Industry Using Goal Programming. Agriculture and Agricultural Science Procedia 3, 154-158.
- Hsueh, C.F., 2015. A bilevel programming model for corporate social responsibility collaboration in sustainable supply chain management. Transportation Research Part E: Logistics and Transportation Review 73, 84-95.
- Hu, H., Lin, T., Liu, Y.Y., Wang, S., Rodríguez, L.F., 2015. CyberGIS-BioScope: a cyberinfrastructure-based spatial decision-making environment for biomass-to-biofuel supply chain optimization. Concurrency and Computation: Practice and Experience 27 (16), 4437-4450.
- Huo, B., Wang, Z., Tian, Y., 2016. The impact of justice on collaborative and opportunistic behaviors in supply chain relationships. Int. J. Prod. Econ. 177, 12-23.
- Hussain, M., Awasthi, A., Tiwari, M.K., 2015. Interpretive structural modeling-analytic network process integrated framework for evaluating sustainable supply chain management alternatives. Applied Mathematical Modelling 40 (5-6), 3671–3687.
- Iacob, S.E., 2015. The Role of the Forest Resources in the Socioeconomic Development of the Rural Areas. Procedia Economics and Finance 23, 1578-1583.
- Jakhar, S.K., 2015. Performance evaluation and a flow allocation decision model for a sustainable supply chain of an apparel industry. J. Clean. Prod. 87, 391-413.
- Jamnadass, R., Langford, K., Anjarwalla, P., Mithöfer, D., 2014. Public-private partnerships in Agroforestry. In: Neal Van Alfen, editor-in-chief. Encyclopedia of Agriculture and Food Systems 4, San Diego: Elsevier, 544-564.

- Jernström, E., Karvonen, V., Kässi, T., Kraslawski, A., Hallikas, J., 2017. The main factors affecting the entry of SMEs into biobased industry. J. Clean. Prod. 141, 1-10.
- Jiang, Z., Wang, F., Duan, Y., Zhou, J., 2016. China's gas market under new situations: Trends and countermeasures Taking Sichuan and Chongqing gas provinces as an example. Natural Gas Industry B 3, 187-194.

Kerr, J., Foster, L., 2011. Sustainable consumption–UK Government activity. Nutrition Bulletin 36 (4), 422-425.

- Klassen, R.D., Vereecke, A., 2012. Social issues in supply chains: Capabilities link responsibility, risk (opportunity), and performance. Int. J. Prod. Econ. 140 (1), 103-115.
- Klassen, R.D., Vachon, S., 2003. Collaboration and evaluation in the supply chain: The impact on plant-level environmental investment. Production and Operations Management 12 (3), 336-352.
- Koh, S.C.L., Gunasekaran, A., Tseng, C.S., 2012. Cross-tier ripple and indirect effects of directives WEEE and RoHS on greening a supply chain. Int. J. Prod. Econ. 140 (1), 305-317.
- Korhonen, J., Pätäri, S., Toppinen, A., Tuppura, A., 2015. The role of environmental regulation in the future competitiveness of the pulp and paper industry: the case of the sulfur emissions directive in Northern Europe. J. Clean. Prod. 108, 864-872.
- Kovács, G., 2008. Corporate environmental responsibility in the supply chain. J. Clean. Prod. 16 (15), 1571-1578.
- Kremer, G.E., Haapala, K., Murat, A. Chinnam, R.B., Kim, K.-Y., Monplaisir, L., Lei, T., 2016. Directions for instilling economic and environmental sustainability across product supply chains. J. Clean. Prod. 112, 2066-2078.
- Kulak, M., Nemecek, T., Frossard, E., Gaillard, G., 2016. Eco-efficiency improvement by using integrative design and life cycle assessment. The case study of alternative bread supply chains in France. J. Clean. Prod. 112, 2452-2461.
- Kumar, S., Malegeant, P., 2006. Strategic alliance in a closed-loop supply chain, a case of manufacturer and eco-non-profit organization. Technovation 26 (10), 1127-1135.
- Kuo, T.C., 2010. The construction of a collaborative-design platform to support waste electrical and electronic equipment recycling. Robotics and Computer-Integrated Manufacturing 26 (1), 100-108.
- Kurdve, M., Shahbazi, S., Wendin, M., Bengtsson, C., Wiktorsson, M., 2015. Waste flow mapping to improve sustainability of waste management: a case study approach. J. Clean. Prod. 98, 304-315.
- Laari, S., Töyli, J., Solakivi, T., Ojala, L., 2016. Firm performance and customer-driven green supply chain management. J. Clean. Prod. 112, 1960-1970.
- Le Dû, A.C., de Corbière, F., 2011. IQ as an enabler of the green and collaborative supply chain. In 16th International conference on Information Quality.
- Leadbitter, D., Benguerel, R., 2014. Sustainable Tuna–Can the Marketplace Improve Fishery Management? Business Strategy and the Environment 23 (6), 417-432.
- Lee, K.H., Kim, J.W., 2011. Integrating suppliers into green product innovation development: an empirical case study in the semiconductor industry. Business Strategy and the Environment 20 (8), 527-538.
- Lehoux, N., D'Amours, S., Langevin, A., 2014. Inter-firm collaborations and supply chain coordination: review of key elements and case study. Production Planning & Control 25 (10), 858-872.
- Leitizia, P., Hendrikse, G., 2016. Supply Chain Structure Incentives for Corporate Social Responsibility: An Incomplete Contracting Analysis. Production and Operations Management 25 (11), 1919-1941.
- Li, A.Q., Found, P., 2016. Lean and Green Supply Chain for the Product-Services System (PSS): The Literature Review and a Conceptual Framework. Proceedia CIRP 47, 162-167.
- Lim, S.J., Phillips, J., 2008. Embedding CSR values: The global footwear industry's evolving governance structure. Journal of Business Ethics 81 (1), 143-156.
- Lin, R.J., 2013. Using fuzzy DEMATEL to evaluate the green supply chain management practices. J. Clean. Prod. 40, 32-39.
- Lind, M., Hägg, M., Siwe, U., Haraldson, S., 2016. Sea Traffic management beneficial for all maritime stakeholders. Transportation Research Procedia 14, 183-192.
- Luo, J., Chong, A.Y.L., Ngai, E.W., Liu, M.J., 2015. Reprint of "Green Supply Chain Collaboration implementation in China: The mediating role of guanxi". Transportation Research Part E: Logistics and Transportation Review 74, 37-49.
- Luzzini, D., Brandon-Jones, E., Brandon-Jones, A., Spina, G., 2015. From sustainability commitment to performance: The role of intra-and inter-firm collaborative capabilities in the upstream supply chain. Int. J. Prod. Econ. 165, 51-63.
- Mahony, G., 2012. Foreign acquisition of agricultural land and food security: a cautionary note on public policy. Economic Papers: A journal of applied economics and policy 31 (4), 501-507.
- Malik, M.M., Abdallah, S., Hussain, M., 2016. Assessing supplier environmental performance: Applying analytical hierarchical process in the United Arab Emirates healthcare chain. Renewable and Sustainable Energy Reviews 55, 1313-1321.
- Maloni, M.J., Benton, W.C., 1997. Supply chain partnerships: opportunities for operations research. European J. Op. Res. 101 (3), 419-429.
- Matt, D.T., Dallasega, P., Rauch, E., 2014. Synchronization of the Manufacturing Process and On-Site Installation in ETO Companies. Procedia CIRP 17, 457-462.
- Mengistie, B.T., Mol, A.P., Oosterveer, P., 2015. Private Environmental Governance in the Ethiopian Pesticide Supply Chain: Importation, Distribution and Use. NJAS-Wageningen Journal of Life Sciences 76, 65-73.
- Mentzer, J., 2001. Supply chain management. Thousand Oaks: SAGE Publications, consulted on 2016-04-15 at: https://books.google.ca/books?id=Y60qVumXKEwC&printsec=frontcover&dq=mentzer+2001+supply+chain&hl=en&sa=X&re dir esc=y#v=onepage&q=joint%20activities&f=false
- Miller, M., Pollard, C., 2005. Health working with industry to promote fruit and vegetables: a case study of the Western Australian Fruit and Vegetable Campaign with reflection on effectiveness of inter-sectoral action. Australian and New Zealand journal of public health 29 (2), 176-182.
- Mirhedayatian, S.M., Azadi, M., Saen, R.F., 2014. A novel network data envelopment analysis model for evaluating green supply chain management. Int. J. Prod. Econ. 147, 544-554.

- Molina-Besch, K., Pålsson, P., 2016. A Supply Chain Perspective on Green Packaging Development-Theory Versus Practice. Packaging Technology and Science 29 (1), 45-63.
- Montoya-Torres, J.R., Ortiz-Vargas, D.A., 2014. Collaboration and information sharing in dyadic supply chains: A literature review over the period 2000–2012. Estudios Gerenciales 30 (133), 343-354.
- Moreno-Peñaranda, R., Gasparatos, A., Stromberg, P., Suwa, A., Pandyaswargo, A.H., de Oliveira, J.A.P., 2015. Sustainable production and consumption of palm oil in Indonesia: What can stakeholder perceptions offer to the debate? Sustainable Production and Consumption 4, 16-35.

Morose, G., Shina, S., Farrell, R., 2011. Supply chain collaboration to achieve toxics use reduction. J. Clean. Prod. 19 (5), 397-407.

- Mourtzis, D., Doukas, M., Psarommatis, F., 2012. Design and planning of decentralised production networks under high product variety demand. Procedia CIRP 3, 293-298.
- Muller, C., Vermeulen, W.J.V., Glasbergen, P., 2012. Pushing or Sharing as Value-driven Strategies for Societal Change in Global Supply Chains: Two Case Studies in the British–South African Fresh Fruit Supply Chain. Business strategy and the environment 21 (2), 127-140.
- Musa, A., Gunasekaran, A., Yusuf, Y., Abdelazim, A., 2014. Embedded devices for supply chain applications: Towards hardware integration of disparate technologies. Expert Systems with Applications 41 (1), 137-155.
- Nakano, K., Hirao, M., 2011. Collaborative activity with business partners for improvement of product environmental performance using LCA. J. Clean. Prod. 19 (11), 1189-1197.
- Nathanail, E., Gogas, M., Adamos, G., 2016. Smart interconnections of interurban and urban freight transport towards achieving sustainable city logistics. Transportation Research Procedia 14, 983-992.
- Nematollahi, M., Hosseini-Motlagh, S.-M., Heydari, J., 2017a. Economic and social collaborative decision-making on visit interval and service level in a two-echelon pharmaceutical supply chain. J. Clean. Prod. 142 (4), 3956-3969.
- Nematollahi, M., Hosseini-Motlagh, S.-M., Heydari, J., 2017b. Coordination of social responsibility and order quantity in a twoechelon supply chain: A collaborative decision-making perspective. Int. J. Prod. Econ. 184, 107-121.
- Newsome, R., Balestrini, C.G., Baum, M.D., Corby, J., Fisher, W., Goodburn, K., Labuza, T.P., Prince, G., Thesmar, H.S., Yiannas, F., 2014. Applications and perceptions of date labeling of food. Comprehensive Reviews in Food Science and Food Safety 13 (4), 745-769.
- Newton, P., Agrawal, A., Wollenberg, L., 2013. Enhancing the sustainability of commodity supply chains in tropical forest and agricultural landscapes. Global Environmental Change 23 (6), 1761-1772.
- Niesten, E., Jolink, A., de Sousa Jabbour, A.B.L., Chappin, M., Lozano, R., 2017. Sustainable collaboration: The impact of governance and institutions on sustainable performance, J. Clean. Prod., In Press, http://dx.doi.org/10.1016/j.jclepro.2016.12.085.
- Nwaka, S., 2005. Drug discovery and beyond: the role of public-private partnerships in improving access to new malaria medicines. Transactions of the Royal Society of Tropical Medicine and Hygiene, 99 (Supplement 1), S20-S29.
- Oelze, N., Hoejmose, S.U., Habisch, A., Millington, A., 2014. Sustainable development in supply chain management: the role of organizational learning for policy implementation. Business Strategy and the Environment 25 (4), 241-260.
- O'Keeffe, J.M., Gilmour, D., Simpson, E., 2016. A network approach to overcoming barriers to market engagement for SMEs in energy efficiency initiatives such as the Green Deal. Energy Policy 97, 582-590.
- Pant, R.R., Prakash, G., Farooquie, J.A., 2015. A Framework for Traceability and Transparency in the Dairy Supply Chain Networks. Procedia-Social and Behavioral Sciences, 189, 385-394.
- Patala, S., Hämäläinen, S., Jalkala, A., Pesonen, H.L., 2014. Towards a broader perspective on the forms of eco-industrial networks. J. Clean. Prod. 82, 166-178.
- Pehlken, A., Madena, K., Aden, C., Klenke, T., 2016. Forming stakeholder alliances to unlock alternative and unused biomass potentials in bioenergy regions. J. Clean. Prod. 110, 66-77.
- Perdikakis, A., Shukla, A., Kiritsis, D., 2015. Optimize Energy Efficiency in the Supply Chain of FMCGs with the Use of Semantic Web Technologies. Procedia Engineering 132, 1112-1119.
- Persson, U., Werner, S., 2012. District heating in sequential energy supply. Applied Energy 95, 123-131.
- Pimenta, H.C., Ball, P.D., 2015. Analysis of Environmental Sustainability Practices Across Upstream Supply Chain Management. Procedia CIRP 26, 677-682.
- Ramanathan, U., Bentley, Y., Pang, G., 2014. The role of collaboration in the UK green supply chains: An exploratory study of the perspectives of suppliers, logistics and retailers. J. Clean. Prod. 70, 231-241.
- Reefke, H., Sundaram, D., 2017. Key themes and research opportunities in sustainable supply chain management identification and evaluation. Omega 66, 195-211.
- Réviron, S., Chappuis, J.M., 2005. Effects of the Swiss retailers' strategy on the governance structure of the fresh food products supply chains. Agribusiness 21 (2), 237-252.
- Richter, B., Bokelmann, W., 2016. Approaches of the German food industry for addressing the issue of food losses. Waste Management 48, 423-429.
- Rizzi, F., Bartolozzi, I., Borghini, A., Frey, M., 2013. Environmental Management of End-of-Life Products: Nine Factors of Sustainability in Collaborative Networks. Business Strategy and the Environment 22 (8), 561-572.
- Rodger, J.A., George, J.A., 2017. Triple bottom line accounting for optimizing natural gas sustainability: A statistical linear programming fuzzy ILOWA optimized sustainment model approach to reducing supply chain global cybersecurity vulnerability through information and communications technology. J. Clean. Prod. 142 (4), 1931-1949.
- Rodríguez, J.A., Gimenez Thomsen, C., Arenas, D., Pagell, M., 2016. NGOS' Initiatives to Enhance Social Sustainability in the Supply Chain: Poverty Alleviation through Supplier Development Programs. J. Supply Chain Management 52 (3), 83-108.
- Roldán, F., Esteban, B., Val, S., 2014. 3D Animation of the Transpyrenean Corridor Created by GIS Tools. Procedia-Social and Behavioral Sciences 160, 244-253.
- Ross, A.D., Jayaraman, V., 2009. Strategic Purchases of Bundled Products in a Health Care Supply Chain Environment. Decision Sciences 40 (2), 269-293.

Roy, R., Whelan, R.C., 1992. Successful recycling through value-chain collaboration. Long range planning 25 (4), 62-71.

- Sahamie, R., Stindt, D., Nuss, C., 2013. Transdisciplinary Research in Sustainable Operations–An Application to Closed-Loop Supply Chains. Business Strategy and the Environment 22 (4), 245-268.
- Sancha, C., Gimenez, C., Sierra, V., 2016. Achieving a socially responsible supply chain through assessment and collaboration. J. Clean. Prod. 112, 1934-1947.
- Sari, K., 2010. Exploring the impacts of radio frequency identification (RFID) technology on supply chain performance. European J. Op. Res. 207 (1), 174-183.
- Schenkel, M., Caniëls, M.C., Krikke, H., van der Laan, E., 2015. Understanding value creation in closed loop supply chains–Past findings and future directions. J. Manufacturing Systems 37, 729-745.
- Schniederjans, D.G., Hales, D.N., 2016. Cloud computing and its impact on economic and environmental performance: A transaction cost economics perspective. Decision Support Systems 86, 73-82.
- Secondi, L., Principato, L., Laureti, T., 2015. Household food waste behaviour in EU-27 countries: A multilevel analysis. Food Policy 56, 25-40.
- Seuring, S., Gold, S., 2013. Sustainability management beyond corporate boundaries: from stakeholders to performance. J. Clean. Prod. 56, 1-6.
- Shanoyan, A., Ross, R.B., Gow, H.R., Peterson, H.C., 2014. Long-term sustainability of third-party facilitated market linkages: Evidence from the USDA marketing assistance program in the Armenian dairy industry. Food Policy 46, 157-164.
- Siddiqui, A.W., Raza, S.A., 2015. Electronic supply chains: Status & perspective. Computers & Industrial Engineering 88, 536-556.
- Sigala, M., 2008. A supply chain management approach for investigating the role of tour operators on sustainable tourism: the case of TUI. J. Clean. Prod. 16 (15), 1589-1599.
- Sime, S.L., 2005. Strengthening the service continuum between transfusion providers and suppliers: enhancing the blood services network. Transfusion 45 (s4), 206S-223S.
- Singh, R.K., Sharma, H.O., Garg, S.K., 2016. Study on Supply Chain Issues in an Auto Component Manufacturing Organization: Case Study. Global Business Review 17 (5), 1196-1210.
- Sippl, K., 2015. Private and civil society governors of mercury pollution from artisanal and small-scale gold mining: A network analytic approach. The Extractive Industries and Society 2 (2), 198-208.
- Smith, C.T., Lattimore, B., Berndes, G., Bentsen, N.S., Dimitriou, I., Langeveld, J.W.A., Thiffault, E., 2016. Opportunities to encourage mobilization of sustainable bioenergy supply chains. WIREs Energy Environ 2016, doi: 10.1002/wene.237.
- Solakivi, T., Laari, S., Toyli, J., Lorentz, H., Ojala, L., 2013. Environmental collaboration in the supply chain and transportation costs of manufacturing and trading companies. 13<sup>th</sup> World Conference on Transport Research, July 15-18.
- Solér, C., Bergström, K., Shanahan, H., 2010. Green supply chains and the missing link between environmental information and practice. Business Strategy and the Environment 19 (1), 14-25.
- Soysal, M., 2016. Closed-loop Inventory Routing Problem for returnable transport items. Transportation Research Part D 48, 31-45.
- Steele, B., Feyerherm, A., 2014. Loblaw Sustainable Seafood: Transforming the Seafood Supply Chain through Network Development and Collaboration, in Christopher G. Worley, Philip H. Mirvis (ed.) Building Networks and Partnerships (Organizing for Sustainable Effectiveness, Volume 3) Emerald Group Publishing Limited 3, 101 – 132.
- Styles, D., Schoenberger, H., Galvez-Martos, J.L., 2012. Environmental improvement of product supply chains: A review of European retailers' performance. Resources, Conservation and Recycling 65, 57-78.
- Tarandi, V., 2015. A BIM Collaboration Lab for Improved through Life Support. Procedia Economics and Finance 21, 383-390.
- Teixeira, A.A., Jabbour, C.J.C., de Sousa Jabbour, A.B.L., Latan, H., de Oliveira, J.H.C., 2016. Green Training and Green Supply Chain Management: Evidence from Brazilian Firms. J. Clean. Prod. 116, 170-176.
- Tesfay, Y.Y., 2014. Environmentally friendly cost efficient and effective sea transport outsourcing strategy: The case of Statoil. Transportation Research Part D: Transport and Environment 31, 135-147.
- Theißen, S., Spinler, S., Huchzermeier, A., 2014. Reducing the Carbon Footprint within Fast-Moving Consumer Goods Supply Chains through Collaboration: The Manufacturers' Perspective. J. Supply Chain Management 50 (4), 44-61.
- Touboulic, A., Walker, H., 2015. Love me, love me not: A nuanced view on collaboration in sustainable supply chains. J. Purchasing & Supply Management 21 (3), 178-191.
- Tranfield, D., Denyer, D., Smart, P., 2003. Towards a methodology for developing evidence-informed management knowledge by means of systematic review. British Journal of Management 14, 207-222.
- Tseng, M.-L., Bui, T.-D., 2017. Identifying eco-innovation in industrial symbiosis under linguistic preferences: A novel hierarchical approach. J. Clean. Prod. 140, 1376-1389.
- Tsoi, J., 2010. Stakeholders' perceptions and future scenarios to improve corporate social responsibility in Hong Kong and Mainland China. Journal of Business Ethics 91 (3), 391-404.
- Türkay, M., Oruç, C., Fujita, K., Asakura, T., 2004. Multi-company collaborative supply chain management with economical and environmental considerations. Computers & Chemical Engineering 28 (6), 985-992.
- Umar, M.S., Jennings, P., Urmee, T., 2013. Strengthening the Palm Oil Biomass Renewable Energy Industry in Malaysia. Renewable Energy 60, 107-115.
- Vachon, S., Klassen, R.D., 2006. Green project partnership in the supply chain: the case of the package printing industry. J. Clean. Prod. 14 (6), 661-671.
- Vachon, S., Klassen, R.D., 2008. Environmental Management and Manufacturing Performance: The Role of Collaboration in the Supply Chain. Int. J. Prod. Econ. 111 (2), 299-315.
- Van Hoof, B.V., Thiell, M., 2014. Collaboration capacity for sustainable supply chain management: small and medium-sized enterprises in Mexico. J. Clean. Prod. 67, 239-248.
- Van Hoof, B.V., Lyon, T.P., 2013. Cleaner production in small firms taking part in Mexico's Sustainable Supplier Program. J. Clean. Prod. 41, 270-282.

- Vrijhoef, R., Koolwijk, J.S.J., Van der Kuij, R.S., Van Oel, C.J., Wamelink, J.W.F., 2014. Developing a monitor for the characterisation of supply chain collaboration and the measurement of its effectiveness in the Dutch social housing sector. In CIB International Conference on Construction in a Changing World, Sri Lanka, India, 4-7 May 2014; Proceedings of the 2014 CIB W55/65/89/92/96/102/117 & TG72/81/83; Authors version. CIB.
- Vurro, C., Russo, A., Perrini, F., 2009. Shaping sustainable value chains: Network determinants of supply chain governance models. J. Business Ethics 90 (4), 607-621.
- Waller, M.A., Fawcett, S.E., Johnson, J.L., 2015. The Luxury Paradox: How Systems Thinking and Supply Chain Collaboration Can Bring Sustainability Into Mainstream Practice. J. Business Logistics 36 (4), 303–305.
- Walker, H., Di Sisto, L., McBain, D., 2008. Drivers and barriers to environmental supply chain management practices: Lessons from the public and private sectors. Journal of Purchasing & Supply Management 14, 69-85.
- Wan Ahmad, W.N.K., Rezaei, J., Tavasszy, L.A., de Brito, M.P., 2016. Commitment to and preparedness for sustainable supply chain management in the oil and gas industry. J. Environmental Management 180, 202-213.
- Wang, G., Gunasekaran, A., Ngai, E.W.T., Papadopoulos, T., 2016. Big data analytics in logistics and supply chain management: Certain investigations for research and applications. Int. J. Prod. Econ. 176, 98-110.
- Wang, Y., Potter, A., Naim, M., Beevor, D., 2011. A case study exploring drivers and implications of collaborative electronic logistics marketplaces. Industrial Marketing Management 40 (4), 612-623.
- Wiengarten, F., Longoni, A., 2015. A nuanced view on supply chain integration: a coordinative and collaborative approach to operational and sustainability performance improvement. Supply Chain Management: An International Journal 20 (2), 139-150.
- Woo, C., Kim, M.G., Chung, Y., Rho, J.J., 2016. Suppliers' communication capability and external green integration for green and financial performance in Korean construction industry. J. Clean. Prod. 112, 483-493.
- Xing, K., Qian, W., Zaman, A.U., 2016. Development of a cloud-based platform for footprint assessment in green supply chain management. J. Clean. Prod. 139, 191-203.
- Yan, M.-R., Chien, K.-M., Yang, T.-N., 2016. Green Component Procurement Collaboration for Improving Supply Chain Management in the High Technology Industries: A Case Study from the Systems Perspective. Sustainability 8, 105.
- Yenipazarli, A., 2017. To collaborate or not to collaborate: Prompting upstream eco-efficient innovation in a supply chain. European J. Op. Res., In Press, http://dx.doi.org/10.1016/j.ejor.2016.12.035.
- Youn, S., Yang, M.G.M., Hong, P., Park, K., 2013. Strategic supply chain partnership, environmental supply chain management practices, and performance outcomes: an empirical study of Korean firms. J. Clean. Prod. 56, 121-130.
- Ytterhus, B.E., Arnestad, P., Lothe, S., 1999. Environmental initiatives in the retailing sector: Analysis of supply chain pressures and partnership, Eco-Management and Auditing 6, 181-188.
- Zander, S., Trang, S., Kolbe, L.M., 2016. Drivers of network governance: a multitheoretic perspective with insights from case studies in the German wood industry. J. Clean. Prod. 110, 109-120.
- Zsidisin, G.A., Siferd, S.P., 2001. Environmental purchasing: a framework for theory development. European J. Purchasing & Supply Management 7 (1), 61-73.

#### APPENDIX A

Table A1 presents the 190 papers that recognize the importance of collaboration in sustainable supply chains. For each, we state the methodology used, the industry targeted, and the categories corresponding to the collaborative mechanisms mentioned.

#### Legend

**RM**: Relationship management

**CEP**: Contractual and economic practices

**JP:** Joint Practices

TISP: Technological and information sharing practices GP: Governance practices

**AP**: Assessment practices

SCD: Supply chain design

Table A1

Author	Methodology Indust	Inductor	Collaborative Mechanism Categories						
Author		industry	RM	СЕР	JP	TISP	GP	AP	SCD
Abbasi et al. (2016)	Quantitative	Automotive	х		х		х		
Acquaye et al. (2015)	Quantitative	Food				х	Х	х	
Adams (2008)	Qualitative	Food					Х		
Aguiar et al. (2013)	Qualitative	Diverse industries	х	х	х	х	х		х
Airike et al. (2016)	Qualitative	Electronic	х	х	х				
Akhavan and Beckmann (2017)	Qualitative	Diverse industries	х	х	х	х	х	х	х
Albino et al. (2012)	Quantitative	Diverse industries	х	х	х				
Amaro and Barbosa-Póvoa (2013)	Modelling	Forest products	х		х				
Amer and Eltawil (2014)	Struct. lit. rev.	N/A	х	х	х				х
Anthony and Ferroni (2012)	Case study	Food	х			х			
Aschemann-Witzel et al. (2017)	Case study	Food	х	х					
Bai et al. (2012)	Modelling	Energy		х	х				
Barari et al. (2012)	Modelling	Apparel	х	х		х	х		
Barnes and Mattsson (2016)	Quantitative	Diverse industries			х	х			
Basiri and Heydari (2017)	Modelling	Retail	х	х	х		х		
Bernstein and Cashore (2007)	Qualitative	N/A	х	х	х		х		
Bishop and Riopelle (2011)	Case study	Chemical	х					х	х
Bisogno (2016)	Qualitative	Food	х						х
Blanco et al. (2016)	Quantitative	Logistics						х	
Blome et al. (2014)	Quantitative	Diverse industries	х		х		х		
Bonou et al. (2016)	Quantitative	Energy						х	х
Boström et al. (2015)	Struct. lit. rev.	N/A	х	х		х	х	х	
Brockhaus et al. (2013)	Qualitative	Diverse industries	х	х	х				х
Brown et al. (2007)	Case study	N/A	х	х		х	х		
Bucklew (2011)	Qualitative	Logistics		х		х			х
Burlingame and Pineiro (2007)	Qualitative	Food			х				
Cao and Zhang (2013)	Modelling	N/A		х	х				
Cao et al. (2015)	Modelling	N/A	х	х			х		
Cao et al. (2017)	Case study	Apparel	х	х	х	Х	х		
Chan et al. (2012)	Quantitative	Diverse industries		х	х	х	х		
Chenga (2011)	Quantitative	Construction	х			х			
Cheung et al. (2009)	Qualitative	Diverse industries	х	х	х	х	х		
Chin et al. (2015)	Qualitative	Manufacturing	х		х	х			
Chkanikova (2015)	Case study	Food	х	х	х		х		
Chopra and Wu (2016)	Quantitative	Electronic			х	х	х	х	
Dai et al. (2015)	Quantitative	Diverse industries	х	х	х	х			
Dale et al. (2016)	Qualitative	Energy	х	х	х	х	х		
Dan et al. (2010)	Quantitative	N/A	х	х	х			х	
Dangelico et al. (2013)	Quantitative	Forest products	х	х		х	х		х
Davidson et al. (2014)	Qualitative	Food	х	х					
De Giovanni and Vinzi (2012)	Quantitative	Diverse industries	х		х		х		
De Giovanni and Vinzi (2014)	Quantitative	Diverse industries	х	х	х	х		х	х
De Sousa Jabbour et al. (2015)	Quantitative	N/A	х		х			х	
De Souza et al. (2014)	Case study	Diverse industries	х	х		х			

Authon	Matha dala m	Talata		Collaborative Mechanism Categories					
Author	Methodology	Industry	RM	CEP	JP	TISP	GP	AP	SCD
Delmotte et al. (2016)	Modelling	Food	х		х	х			
Ding et al. (2015)	Modelling	Automotive		х	х		х		
Ding <i>et al.</i> (2016a)	Modelling	Automotive	х	х			х		х
Ding et al. (2016b)	Modelling	Automotive		х			х		х
Dissanayake and Sinha (2015)	Qualitative	Manufacturing	х		х			х	х
Egels-Zandén et al. (2015)	Case study	Apparel	х	х	х		х		
Ellram and Golicic (2016)	Case study	Logistics	х	х		х	х	х	
Fiorino and Bhan (2014)	Qualitative	Electronic	х	х	х	х	х	х	
Fischer et al. (2016)	Qualitative	Automotive	х		х	х		х	
Fischer and Pascucci (2017)	Case study	Apparel	х	х	х	х	х		х
Flint and Signori (2014)	Qualitative	Food	х		х	х			
Formentini and Taticchi (2016)	Case study	Diverse industries	х	х	х	х	х	х	
Forsman et al. (2014)	Qualitative	Food	х	х	х		х	х	х
Gallear et al. (2015)	Quantitative	Diverse industries	х	х	х	х	х	х	
Gavronski et al. (2011)	Quantitative	Manufacturing	х		х	х		х	х
Germani et al. (2015)	Case study	Manufacturing	х			х			х
Ghosh and Shah (2015)	Modelling	N/A	х	х	х				
Gimenez et al. (2012)	Quantitative	Diverse industries			х		х	х	
Gold et al. (2010)	Struct. lit. rev.	N/A	х	х	х	х	х		
Govindan et al. (2013)	Modelling	N/A	х		х				х
Goyal et al. (2017)	Case study	Diverse industries	х			х	х		х
Grekova et al. (2016)	Quantitative	Food	х	х	х	х			
Grimm et al. (2016)	Case study	Diverse industries	х	х	х	х		х	
Gunasekaran et al. (2015)	Struct. lit. rev.	N/A	х	х	х				
Hajdul (2014)	Modelling	Diverse industries			х	х			
Hall et al. (2013)	Qualitative	Logistics	х	х		х	х		
Halloran et al. (2014)	Case study	Food	х	х	х	х	х	х	
Hazelton et al. (2013)	Qualitative	Manufacturing	х	х		х	х		
Hisjam et al. (2015)	Modelling	Forest products		х					
Hsueh (2015)	Modelling	N/A	х	х		х	х		
Hu et al. (2015)	Modelling	N/A	х			х			
Huo et al. (2016)	Quantitative	Diverse industries	х	х					
Hussain et al. (2015)	Modelling	N/A	х	х	х	х	х	х	
Iacob (2015)	Qualitative	Forest products		х			х		
Jakhar (2015)	Modelling	Apparel	х	х	х		х		х
Jamnadass et al. (2014)	Case study	Food		х	х	х	х	х	
Jernström et al. (2017)	Quantitative	Forest products							
Jiang et al. (2016)	Qualitative	Oil and gas	х	х	х	х	х		х
Kerr and Foster (2011)	Qualitative	Food			х	х	х		
Klassen and Vachon (2003)	Quantitative	Manufacturing	х	х	х	х	х		
Klassen and Vereecke (2012)	Case study	Diverse industries	х	х	х	х	х	х	
Koh et al. (2012)	Case study	Information Tech.	х				х		
Korhonen et al. (2015)	Qualitative	Forest products	х		х			х	
Kovács (2008)	Case study	Diverse industries	х		х			х	
Kremer et al. (2016)	Modelling	Manufacturing							
Kulak et al. (2016)	Case study	Food	х	х	х				
Kumar and Malegeant (2006)	Case study	Apparel	х			х			
Kuo (2010)	Modelling	Electronic	х			х			
Kurdve et al. (2015)	Case study	Manufacturing	х	x	Х	х	х	x	х
Laari et al. (2016)	Quantitative	Manufacturing	х	x	Х	х	х	х	
Le Dû and de Corbière (2011)	Qualitative	Retailer sector				х	х		х
Leadbitter and Benguerel (2014)	Qualitative	Food	х	x	Х		х	х	
Lee and Kim (2011)	Case study	Electronic	х	х	Х		х		
Leitizia and Hendrikse (2016)	Modelling	Food	х	x					x
Li and Found (2016)	Qualitative	N/A	х	х	Х	х	х	х	
Lim and Phillips (2008)	Case study	Apparel		х	х	х	Х		

	Malada	<b>.</b>		Colla	itegories				
Author	Methodology	Industry	RM	CEP	JP	TISP	GP	AP	SCD
Lin (2013)	Modelling	Electronic	х	х	х				
Lind et al. (2016)	Qualitative	Logistics	х	х	х	х	х	х	х
Luo et al. (2015)	Quantitative	Manufacturing	х	х	х	х		х	
Luzzini et al. (2015)	Quantitative	Diverse industries	х	х		х	х	х	х
Mahony (2012)	Case study	Food	х				х		
Malik et al. (2016)	Case study	Healthcare	х	х	х	х	х		х
Maloni and Benton (1997)	Struct. lit. rev.	N/A	х	x	х	х		х	х
Matt <i>et al.</i> (2014)	Case study	Construction	х		х		х		
Mengistie <i>et al.</i> (2015)	Qualitative	Food		x	х	х	х		
Miller and Pollard (2005)	Case study	Food	х	x	х	x			
Mirhedavatian <i>et al.</i> (2014)	Modelling	Food							
Molina-Besch and Pålsson (2016)	Case study	Diverse industries	x	x			x		
Moreno-Peñaranda <i>et al.</i> (2015)	Case study	Oil and gas	v	~	v		~		
Morose <i>et al.</i> (2011)	Case study	Electronic	А		А	v			
Molose et al. (2011)	Modelling	Automotive				A V	v		v
Mullor et al. (2012)	Case study	Food	v	v	v	х	X		х
Muller et al. (2012)	Case study	FOOD	X	X	X		X		
Musa et al. (2014)	Struct. Int. rev.	N/A Discourse in Association	X	X	X	X			
		Diverse industries	X	X	v	X	v	v	v
Nathanail <i>et al.</i> (2016)	Quantitative	Logistics	X	X	X	X	X	X	X
Nematollahi <i>et al.</i> (2017a)	Modelling	Pharmaceutical			X			X	
Nematollahi <i>et al.</i> (2017b)	Modelling	N/A		X	X			Х	
Newsome <i>et al.</i> (2014)	Qualitative	Food	х			X	X		
Newton <i>et al.</i> (2013)	Struct. lit. rev.	Food		х		X	X		
Niesten <i>et al.</i> (2017)	Qualitative	N/A	х				х		
Nwaka (2005)	Case study	Pharmaceutical	х	Х	х				
Oelze <i>et al.</i> (2014)	Qualitative	Diverse industries	х	Х	х	х			
O'Keeffe et al. (2016)	Case study	Construction	х	х		х	х	х	
Pant et al. (2015)	Case study	Food	х	x	х	х	х		
Patala et al. (2014)	Struct. lit. rev.	N/A	х	х	х	х	х		
Pehlken et al. (2016)	Qualitative	Energy	х	х					х
Perdikakis et al. (2015)	Case study	N/A	х			х	х		
Persson and Werner (2012)	Quantitative	Energy	х	х	х				
Pimenta and Ball (2015)	Struct. lit. rev.	N/A	х		х	х	х	х	
Ramanathan et al. (2014)	Case study	Diverse industries	х	X	х	х			
Reefke and Sundaram (2017)	Qualitative	Diverse industries	х	х	х	х	х	х	х
Réviron and Chappuis (2005)	Case study	Food	х	х	х		х	х	
Richter and Bokelmann (2016)	Quantitative	Food	х	x		х			
Rizzi et al. (2013)	Struct. lit. rev.	N/A	х	X	х	x	х		
Rodger and George (2017)	Modelling	Energy		x		х	х		
Rodríguez <i>et al.</i> (2016)	Case study	Diverse industries	х	х	х	х	х		х
Roldán et al. $(2014)$	Modelling	Logistics	х	x		x			
Ross and Javaraman (2009)	Modelling	Healthcare		x					x
Roy and Whelan (1992)	Case study	Electronic	х	X	х	x	х		
Sahamie <i>et al.</i> (2013)	Struct lit rev	Diverse industries	x	x	x	x			
Sancha <i>et al.</i> (2016)	Quantitative	Diverse industries	v		v			v	
Sari (2010)	Modelling		л v		x	x		^	
Sahankal et al. (2015)	Struct lit roy	N/A N/A	A X	v	x	v	v		
Scheinker et ut. (2013)	Ouentitative	IN/A	А	A		л 			
Secondi et al. (2015)	Quantitative	Food	v	v		X			
Second <i>et al.</i> (2015)	Quantitative		<u>л</u>	А		X	v	v	v
Seuring and Gold (2013)	Qualitative	IN/A	X		х	<b> </b>	X	X	X
Shanoyan <i>et al.</i> (2014)	Quantitative	Food		X			X		
Siddiqui and Raza (2015)	Struct. lit. rev.	N/A	X		X	X	X	X	X
Sigala (2008)	Case study	Tourism		х	х	X	х	х	х
Sime (2005)	Qualitative	Healthcare	х	x	х		х		
Singh <i>et al.</i> (2016)	Case study	Automotive	х	X	х	X		Х	
Sippl (2015)	Qualitative	Mining	х	Х	Х	Х	Х		

Authon	Methodology Ir	Industry	Collaborative Mechanism Categories						
Author			RM	CEP	JP	TISP	GP	AP	SCD
Smith et al. (2016)	Qualitative	Energy	х	х		х	х		х
Solakivi et al. (2013)	Quantitative	Diverse industries	х		х				
Solér et al. (2010)	Case study	Food	х	х	х	х	х		
Soysal (2016)	Modelling	Food						х	
Steele and Feyerherm (2014)	Case study	Food		х	х	х			х
Styles et al. (2012)	Quantitative	Retailer sector	х				х		х
Tarandi (2015)	Qualitative	Construction	х	х		х	х		
Teixeira et al. (2016)	Quantitative	Manufacturing	х		х		х		
Tesfay (2014)	Case study	Oil and gas	х	х		х	х	х	х
Theiβen et al. (2015)	Case study	Manufacturing	х	х	х	х	х		
Touboulic and Walker (2015)	Qualitative	Food	х	х	х	х	х	х	
Tseng and Bui (2017)	Quantitative	Apparel			х	х	х	х	х
Tsoi (2010)	Case study	Apparel	х	х	х		х		
Türkay et al. (2004)	Modelling	Energy			х			х	
Umar et al. (2013)	Case study	Oil and gas		х					
Vachon and Klassen (2006)	Quantitative	Package printing	х	х	х				
Vachon and Klassen (2008)	Quantitative	Manufacturing	х		х				
Van Hoof and Lyon (2013)	Quantitative	Manufacturing	х	х		х			
Van Hoof and Thiell (2014)	Quantitative	Diverse industries	х	х	х	х			
Vrijhoef et al. (2014)	Qualitative	Construction	х	х	х	х	х		х
Vurro et al. (2009)	Qualitative	N/A	х	х	х		х		
Waller et al. (2015)	Qualitative	N/A				х		х	
Wan Ahmad et al. (2016)	Qualitative	Energy	х		х	х	х	х	х
Wang et al. (2011)	Case study	Logistics	х	х		х	х		
Wang et al. (2016)	Qualitative	N/A	х			х	х	х	
Wiengarten and Longoni (2015)	Quantitative	Manufacturing	х	х	х	х	х		
Woo et al. (2016)	Quantitative	Construction	х	х	х	х			
Xing et al. (2016)	Qualitative	Apparel				х		х	
Yan et al. (2016)	Quantitative	Electronic	х	х	х		х	х	
Yenipazarli (2017)	Modelling	N/A	х	х	х				
Youn et al. (2013)	Quantitative	Manufacturing	х	х		х			
Ytterhus et al. (1999)	Qualitative	Food	х	х	х	х	х		
Zander et al. (2016)	Case study	Forest products	х	х	х	х	х		
Zsidisin and Siferd (2001)	Struct. lit. rev.	N/A	Х	х	х	х	х	х	